Author's response to reviews

Title: Low-molecular-weight heparin for prevention of placenta-mediated pregnancy complications: protocol for a systematic review and individual patient data meta-analysis (AFFIRM)

Authors:

Marc A Rodger (mrodger@ohri.ca)
Nicole J Langlois (nlanglois@ohri.ca)
Johanna IP de Vries (jip.deVries@vumc.nl)
Evelyne Rey (evelyne_rey@ssss.gouv.qc.ca)
Ida Martinelli (martin@policlinico.mi.it)
Ekkehard Schleussner (ekkehard.schleussner@med.uni_jena.de)
Timothy Ramsay (tramsay@ohri.ca)
Ranjeeta Mallick (rmallick@ohri.ca)
Becky Skidmore (bskidmore@rogers.com)
Saskia Middeldorp (s.middeldorp@amc.uva.nl)
Shannon Bates (tabesm@mcmaster.ca)
David Petroff (david.petroff@zks.uni-leipzig.de)
Dick Bezemer (pdickbezemer@planet.nl)
Marion E van Hoorn (m.vanhoorn@vumc.nl)
Carolien NH Abheiden (c.abheiden@vumc.nl)
Analisa Perna (annalisa.perna@marionegri.it)
Paulien de Jong (p.q.dejong@amc.uva.nl)
Risto Kaaja (risto.kaaja@hus.fi)

Version: 3
Date: 1 April 2014

Author's response to reviews:

The Acknowledgement section has been updated. The names and contributions of collaborators who have given their permission to be acknowledged have been added.

One of the secondary outcomes (venous thromboembolism) was inadvertently omitted from the previous draft and has been added as outcome #19.